Author’s response to reviews

Title: Factors associated with early introduction of complementary feeding and consumption of non-recommended foods among Dutch infants: the BeeBOFT study

Authors:
Lu Wang (l.wang@erasmusmc.nl)
Amy van Grieken (a.vangrieken@erasmusmc.nl)
Laura van der Velde (L.A.van_der_Velde@lumc.nl)
Eline Vlasblom (eline.vlasblom@tno.nl)
Maaike Beltman (maaike.beltman@tno.nl)
Monique L’Hoir (M.LHoir@ggdnog.nl)
Magda Boere-Boonekamp (m.m.boere-boonekamp@utwente.nl)
Hein Raat (h.raat@erasmusmc.nl)

Version: 3 Date: 20 Mar 2019

Author’s response to reviews:

Dear Editor,

Please find the final version of our manuscript (PUBH-D-18-01404) entitled “Factors associated with early introduction of complementary feeding and consumption of non-recommended foods among Dutch infants: the BeeBOFT study”. We would like to thank the editor and reviewer for thoroughly evaluating our manuscript again. Each comment made by the editor has been addressed. We confirm that the submitted manuscript is the final clean version with no tracked changes, and all the tables and figures are clean versions. We have corrected the email address for our co-author Dr. Magda M Boere-Boonekamp. We have added an ‘Acknowledgement’ section and moved the sentence ‘Professional advice on the English was given by J. Burrough’ to this section. We have corrected the Trial registration part as requested. Further, we have included the questionnaires measuring infant feeding we used in the present study as supplemental file. We confirm that we have added references for the other questionnaires we have used in the present study when necessary (see line 226, line 231, line 238). We hope we have adequately addresses the editor’s comments and suggestions and we look forward to your response.
Sincerely,

Hein Raat,
On behalf of all coauthors